Author's response to reviews

Title: Factors Associated with Default from Treatment among Tuberculosis Patients in Nairobi Province, Kenya: A Case Control Study

Authors:

Bernard N Muture (bnmuture@yahoo.com)
Margaret N Keraka (mnychoka2000@yahoo.com)
Peter K Kimuu (pkimuu@yahoo.com)
Ephantus W Kabiru (ewkabiru@yahoo.com)
Victor Ombeka (ombeka@yahoo.com)
Francis Oguya (foguya@yahoo.com.au)

Version: 11 Date: 2 September 2011

Author's response to reviews:

Revision was minimum but a few typo errors were corrected in the main text. One of the authors address (NLTP) was corrected. Tables and figures were ok and no revisions were made on them

Thank you

Muture